Author's response to reviews

Title: Long term follow up results of sequential left internal thoracic artery graft on severe left anterior descending artery disease

Authors:

Murat Mert (mmert@superonline.com)
Gurkan Cetin (gurkancetin@hotmail.com)
Cenk E Yildiz (cenksesr@yahoo.com)
Murat Ugurlucan (muratugurlucan@yahoo.com)
Ilker M Çaglar (ilkermuratcaglar@gmail.com)
Ahmet Ozkara (ahozmetkar@yahoo.com)
Atif Akcevin (aaakcev@yahoo.com)
Cihat Bakay (cihbak67@yahoo.com)

Version: 3 Date: 27 September 2010

Author's response to reviews:

Dear Editor,

We received your and reviewers comments. The manuscript has been revised accordingly and changes performed on the text are highlighted as well as summarized below.

We are looking forward to your comments and decision on our paper.

Yours sincerely.
Murat Ugurlucan, MD.

Revisions:

Reviewer 1:
1- In the discussion section, the paragraph before conclusion is deleted.
2- The references 19-21 are deleted.
3- Extensive copy-editing and correction of writing are performed by a native speaker.

Reviewer 2:
1- An additional figure is added.
2- Language errors are corrected by a native speaker.

Reviewer 3:
1- Language errors are corrected by a native speaker.